Author's response to reviews

**Title:** Exploring the relationships between International Classification of Functioning, Disability and Health (ICF) constructs in people with osteoarthritis.

**Authors:**

- Beth Pollard (beth.pollard@abdn.ac.uk)
- Marie Johnston (m.johnston@abdn.ac.uk)
- Paul Dieppe (paul.dieppe@pms.ac.uk)

**Version:** 2  **Date:** 19 October 2010

**Author's response to reviews:**

We have added more detail to the results section of the abstract as requested. We have also added the ethics approval information to the methods section of the manuscript together with the declaration of the research conforming to the helsinki declaration.